Author’s response to reviews

Title: Propensity score matching analysis of a phase II study on simultaneous modulated accelerated radiation therapy using helical tomotherapy technique in patients with nasopharyngeal carcinoma

Authors:
Lei Du (coolbile_1999@163.com)
Xinxin Zhang (xinxinzhang66@hotmail.com)
Linchun Feng (301flc@163.com)
Baolin Qu (qubl6212@sina.com)
Jing Chen (chenjing19861206@163.com)
Jun Yang (yang74008@163.com)
Haixia Liu (crystal0214@163.com)
Shouping Xu (xshp228@163.com)
Chuanbin Xie (xiechuanbin2003@163.com)
Lin Ma (malinpharm@sina.com)

Version: 4 Date: 03 Aug 2017

Author’s response to reviews:

Editor Comments:
As we are unable to find the data from the url: http://www.medresman.org/pub/en/proj/listbyproj.aspx?proj=627, please include a statement such as: 'The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.' in the Availability of data and materials section or include a link that takes readers to the data

Response: The statement 'The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.' has been indicated in the Availability of data and materials section.